Author's response to reviews

Title: A study of the number of physicians registered in Jamaica and their countries of training.

Authors:

Jennifer Knight-Madden (jennifer_knight@yahoo.com)
Robert Gray (rhgray@cwjamaica.com)

Version: 7 Date: 8 December 2008

Author's response to reviews:

The revisions are very minor and are final. I have added the words of Jamaica after Paediatric Association making it now Paediatric Association of Jamaica (lines 7-8, page 6).

I have also deleted an empty line in the lowermost box of figure 1 which had obscured from view the total number of physicians presumably in practice.